Author's response to reviews

Title: Post mortem magnetic resonance imaging in the fetus, infant and child: A comparative study with conventional autopsy (MaRIAS Protocol)

Authors:

Sudhin Thayyil Dr (s.thayyil@ucl.ac.uk)
Neil J Sebire Prof (SebirN@gosh.nhs.uk)
Lyn S Chitty Prof (l.chitty@ich.ucl.ac.uk)
Angie Wade Dr (a.wade@ich.ucl.ac.uk)
Oystein Olsen Dr (OlsenO@gosh.nhs.uk)
Roxana Gunny Dr (GunnyR1@gosh.nhs.uk)
Amaka Offiah Dr (amaka.offiah@nhs.net)
Dawn E Saunders Dr (SaundD@gosh.nhs.uk)
Catherine M Owens Dr (OWENSC@gosh.nhs.uk)
WK Kling Chong Dr (Kling.Chong@gosh.nhs.uk)
Nicola J Robertson Dr (n.robertson@ucl.ac.uk)
Andrew M Taylor Prof (a.taylor76@ucl.ac.uk)

Version: 2 Date: 25 October 2011

Author's response to reviews:

Dear Editor,

Please find attached the evidence of external funding (UK Department of Health) and ethics approval. No changes have been made to the manuscript.

Best Wishes

Dr Sudhin Thayyil